Author's response to reviews

Title: Analysis of the association between paternity and reoperation for urethral obstruction in adult hypospadias patients who underwent two-stage repair in childhood

Authors:

Akihiro Kanematsu (aqui@hyo-med.ac.jp)
Shiro Tanaka (tanaka.shiro.8n@kyoto-u.ac.jp)
Takahiko Hashimoto (th-uro@hyo-med.ac.jp)
Michio Nojima (nojimam@hyo-med.ac.jp)
Shingo Yamamoto (shingo@hyo-med.ac.jp)

Version: 1 Date: 09 Jan 2019

Author's response to reviews:

Editor's comments:

1. Identifiable information

Please note that identifying/confidential patient data should not be shared. For example, age, sex, ethnicity and religion should be presented as summaries only (E.g. x females, y males, age range a-b, median c, etc). As the Supplementary table 1 in the current manuscript contains individual details, please revise the relevant information accordingly to avoid compromising anonymity or consent to publish must be obtained from these people, or in the case of children, their parent or legal guardian to publish these potentially identifiable data if these were intended to be included in your submission.

We deleted supplementary table, and according descriptions in the manuscript.

2. Declarations formatting

Please also include a 'Consent for publish' and 'Acknowledgements' subsection in the Declarations according to the submission guidelines
https://bmcurol.biomedcentral.com/submission-guidelines/preparing-your-manuscript/research-article
We included 'Consent to publish' (not applicable because we deleted the supplementary Table). Thank you for pointing out to include Acknowledgments, otherwise we should have had omitted important contributions.

3. Section heading

Please rename the section heading 'Introduction' to 'Background' in the main text of this manuscript to ensure it is in line with journal requirements.

Revised accordingly.